RESEARCH

Open Access



Validation of the Musculoskeletal Health Questionnaire in a general population sample: a cross-sectional online survey in Hungary

Áron Hölgyesi^{1,2}, Gyula Poór³, Petra Baji⁴, Zsombor Zrubka^{5,6}, Miklós Farkas⁷, Ágota Dobos⁸, László Gulácsi^{5,6}, Levente Kovács⁹ and Márta Péntek^{5*}

Abstract

Background: The Versus Arthritis Musculoskeletal Health Questionnaire (MSK-HQ) measures symptom severity and health-related quality of life (HRQoL) of people with musculoskeletal (MSK) conditions. We aimed to test the psy-chometric properties of the MSK-HQ among the general adult population and identify the determinants of MSK-HQ states. In addition, we aimed to explore the relationship between MSK-HQ and standard well-being measurement tools.

Methods: The translation process of the MSK-HQ into Hungarian followed the standard methods provided by the developer. A cross-sectional online survey was performed in Hungary involving a population normative sample (N = 2004, women: 53.1%; mean age: 48.3, SD = 16.6 years). Socio-demographic characteristics and self-reported MSK disorders were recorded. Alongside the MSK-HQ, standard measures of HRQoL (EQ-5D-5L), physical functioning (HAQ-DI) and well-being (ICECAP-A/O, WHO-5, Happiness VAS) were applied. Clinical and convergent validity were assessed by subgroup comparisons (Mann–Whitney-U and Kruskal–Wallis tests) and Spearman's rank correlations. Internal consistency was assessed by Cronbach's alpha. Test–retest reliability (N = 50) was evaluated by intraclass correlation coefficient (ICC). Predictors of MSK-HQ were analysed by ordinary least square multiple regressions.

Results: The mean MSK-HQ index score was 44.1 (SD = 9.9). MSK-HQ scores were significantly lower in subgroups with self-reported MSK disorders. Correlations were strong between MSK-HQ and EQ-5D-5L (0.788), EQ VAS (0.644) and HAQ-DI (-0.698) and moderate with the well-being measures (p < 0.05). Cronbach's alpha was 0.924 and ICC was 0.936 (p < 0.05). Being a man, living in the capital, having higher income and education were positively associated with MSK-HQ scores.

Conclusions: This is the first study to prove the validity and reliability of the MSK-HQ among the general public. The impact of socio-demographic characteristics on MSK-HQ scores deserves consideration in clinical studies.

Keywords: Musculoskeletal health, Quality of life, Well-being, Patient reported outcome measures, Validation, Psychometrics, Population sample, Hungary

*Correspondence: pentek.marta@uni-obuda.hu

⁵ Health Economics Research Center, University Research and Innovation Center, Óbuda University, Budapest, Hungary Full list of author information is available at the end of the article



Plain English summary

Musculoskeletal (MSK) disorders are among the most common chronic diseases affecting approximately 1.71 billion people worldwide and one in four people in Europe with an increasing prevalence by age. The Versus Arthritis Musculoskeletal Health Questionnaire

© The Author(s) 2022. **Open Access** This article is licensed under a Creative Commons Attribution 4.0 International License, which permits use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons licence, and indicate if changes were made. The images or other third party material in this article are included in the article's Creative Commons licence, unless indicated otherwise in a credit line to the material. If material is not included in the article's Creative Commons licence and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder. To view a copy of this licence, wisit http://creativecommons.org/licenses/by/4.0/. The Creative Commons Public Domain Dedication waiver (http://creativecommons.org/publicdomain/zero/1.0/) applies to the data made available in this article, unless otherwise stated in a credit line to the data.

(MSK-HQ) was developed to assess symptoms and quality of life of people with diverse MSK conditions (such as arthritis or back pain) in a standardised and comparable way. The measurement properties of the MSK-HQ have been proven in various MSK patient samples. However, little is known about its performance among the general public and how differences in socio-demographic status (e.g. age, sex, residency, educational level) of the respondents may influence the results. In our study we addressed these two main issues. We have produced and validated the Hungarian version of the MSK-HQ in an online survey involving a representative sample of the adult Hungarian population. Results of the MSK-HQ were compared against respondents' health status, MSK problems and standard measures of health-related quality of life, physical functioning and wellbeing. Statistical analyses followed international standards. Results confirmed that MSK-HQ is a credible measurement tool for the assessment of MSK health-related quality of life of the general public. Certain socio-demographic factors (sex, residency, educational and income level) were shown to have significant impact on MSK-HQ results that should be considered in both the design of and conclusions drawn from MSK-HQ clinical studies.

Background

Musculoskeletal (MSK) disorders are among the most common chronic diseases world wide that may have a negative impact on patients' physical functioning, healthrelated quality of life (HRQoL) and ability to work, causing significant social and economic burden [1, 2].

The increasing use of generic HRQoL measures (such as the EQ-5D-5L, SF-36) can be observed in MSK diseases both in clinical trials and practice, however these tools do not specifically capture MSK health-related problems [3]. There has therefore been demand for an instrument which focuses on the impact of MSK disorders on patients' life that can be used across different MSK conditions thus allowing for the comprehensive assessment of overall MSK health [4-7]. The Versus Arthritis Musculoskeletal Health Questionnaire (MSK-HQ) has been developed with the intent to fill this gap as it incorporates the assessment of patients' MSK related symptoms and HRQoL aspects in one questionnaire and can be used in various MSK conditions [8]. The MSK-HQ also enables monitoring patients with MSK problems over time and makes it possible to compare the performance of health services utilized due to MSK disorders [8–10]. The MSK-HQ has been validated in selected patient samples with a wide range of MSK diagnosis and a number of language versions have become available as well [8, 9, 11-18]. However, the MSK-HQ has not been tested in the general population so far, therefore population normative data are not available and little is known about the impact of socio-demographic factors on MSK-HQ results.

In addition, most MSK-HQ validation studies applied only the EQ-5D-5L generic HRQoL measure and disease-specific tools as reference standards. The relationship between MSK-HQ and well-being measures has remained unexplored so far, while the assessment of the impact on well-being is receiving increasing attention in medical decision making [19, 20], health economic evaluations [21, 22], policy making [23] and in developing social care interventions [24]. The validation of the Hungarian language version would allow us to contribute to MSK-HQ research by targeting these questions, and also to broaden the use of the MSK-HQ instrument to Hungary.

The focus of this study is two-fold. Primarily, we aimed to establish and validate the Hungarian version of the MSK-HQ, analyse its psychometric properties among the general population and evaluate the effects of sociodemographic factors on MSK-HQ outcomes. Secondarily, we aimed to assess the associations between MSK-HQ and commonly used well-being measures.

Methods

Data collection

A cross-sectional online survey was conducted in May-June 2020 among the general adult population (age \geq 18) of Hungary. Recruitment of respondents and data collection were performed by a survey company (New Land Media Kft. - Századvég) involving members of an online access panel. Details on participation rate were confidential. The company was responsible also for data protection and compliance with GDPR. Anonymized data were handled for the analyses. Quota sampling (a non-probability sampling method) was used in order to obtain a representative sample of the normal adult population. Quotas were set for age, sex, educational level and type of settlement according to the 2011 Population Census [25]. The targeted sample size was 2000, which is the double of the usually needed sample size (N=1000) to achieve representativeness defined by the above detailed criteria. Nonetheless, we aimed to achieve a larger than a minimum sample size in all age-groups and a remarkable number of participants with MSK problems. Ethical approval was obtained from the Hungarian Medical Research Council (no. IV/565-5/2020/EKU). Respondents were informed that participation in the survey was voluntary, their data would remain anonymous, impersonal and would be used solely for scientific purposes. Before starting the survey, informed consent was obtained from all individual participants included in the study.

The questionnaire

The survey questionnaire consisted of two modules. The first module (reported in this paper) focused on MSK health, the second one assessed respondents' subjective expectations towards the length and quality of their life. Socio-demographic characteristics of the sample was recorded. Participants completed the Hungarian version of MSK-HQ along with standard outcome measures of health status, HRQoL, physical functioning and well-being. MSK health problems and diagnoses were explored with relevant questions of the European Health Interview Survey (EHIS) [22]. Respondents were asked to indicate if they used healthcare services (hospital admission, specialist care, general practitioner) or received informal care due to MSK problems. Informal caregivers were also identified by self-reports regardless of MSK health status. To assess test-retest reliability, the completion of MSK-HQ was repeated in the end of the survey in a subgroup of 50 respondents who were selected randomly from the study sample.

Outcome measures

MSK-HQ and its translation and validation for Hungary

The MSK-HQ was developed in 2016 as a specific outcome measure to be used by people with different MSK conditions [8]. The questionnaire assesses how much respondents' MSK problems have affected their life in the last two weeks. It consists of 14 items, from which eleven measure symptoms and HRQoL, while three focus on patients' attitude towards their condition and the overall impact of symptoms. Response options are operationalized on a five-point Likert scale, ranging from 4 to 0 ('not at all' and 'extremely', respectively), except items 12 and 13, which are given in reverse order. The final score, ranging from 0 to 56, is calculated by summing up the scores given for each item. Higher scores indicate better MSK health status. An additional, fifteenth item assesses the number of days with significant physical activity in the past week but this item is not included in the final score.

The validation of the Hungarian version was carried out in accordance with the protocol provided by the developer of the original MSK-HQ (Oxford University Innovation). In brief, forward translations were performed independently by three researchers. A reconciled Hungarian version was produced via discussions which was then back translated independently by two experts. Results were reviewed by the developer and items with uncertanities went through a second round forward-back translation cycle. Cognitive debriefing was performed involving five patients with diverse MSK problems. After proofreading and quality check, the final Hungarian version was accepted as the best available language version by the developer.

МЕНМ

The Minimum European Health Module (MEHM) is composed of three questions concerning three different aspects of health: self-percieved health (response options: very good, good, fair, bad, very bad), long standing illness or chronic morbidity (response options: yes or no) and the presence of long-standing activity limitations measured by the Global Activity Limitation Indicator (GALI) (reponse options: severely limited, limited but not severely, not limited at all) [26].

EQ-5D-5L

The EQ-5D-5L is a generic questionnaire that evaluates HRQoL by assessing the following five domains: mobility, self-care, usual activities, pain/discomfort, anxiety/ depression [27]. Respondents are asked to indicate their current health state on a five-level Likert-scale (ranging from 1 –'no problems' to 5 –'unable to'). To calculate EQ-5D-5L index score, we used the Hungarian tariffs (range: -0.848 – 1.000) [28]. EQ VAS is an additional item that measures respondents self-reported health on a visual analogue scale, ranging from 0 to 100, where 0 indicates the worst and 100 indicates the best health status that the respondent can imagine. Higher scores indicate better health status.

HAQ-DI

Health Assessment Questionnaire – Disability Index (HAQ-DI) [29] measures functional ability over the past week with 20 items across the following 8 domains: dressing, arising, eating, walking, hygiene, reach, grip and common activities. Each domain contains 2 or 3 items. Possible response options for each item range from 0 to 3 (0—'without difficulty' and 3—'unable to do,' respectively). Eighteen additional items are provided to indicate if aids or assisstive devices were used for the activities listed in the domains. To calculate the HAQ-DI score, we applied the alternative scoring method (no correction for aid or devices was made) thus the highest item score within each domain was considered to calculate the total score (range 0–3, higher score indicates worse status).

ICECAP-A and ICECAP-O

The ICECAP-A and ICECAP-O tools were developed to evaluate capability well-being of adults (18 +) and older people (65 +), respectively [30, 31]. Each version contains five items. For ICECAP-A these are attachment, stability, achievement, enjoyment and autonomy, while ICECAP-O items are attachment, security, role, enjoyment and control. Respondents are asked to indicate their current capability well-being on a 4-level scale (ranging from 4—'full capability' to 1—'no capability'). Index scores for each state can be calculated by using tariffs obtained

WHO-5

The World Health Organisation-Five Well-Being Index (WHO-5) evaluates self-reported mental well-being in relation to the last two weeks [34]. The questionnaire comprises five items. Response options are available on a six-level scale (ranging from 0—'at no time' to 5—'all of the time'). The final score is calculated by the multiplication of the sum of item scores by 4. Higher scores indicate better well-being.

Happiness VAS

A visual analogue scale (VAS) of happiness was applied to measure the current subjective degree of happiness of respondents (where 0 indicates 'completely unhappy' and 10 indicates 'completely happy').

Statistical analysis

Socio-demographic characteristics and health status of the sample were analyzed using descriptive statistics. The psychometric characteristics of MSK-HQ were assessed in relation to socio-demographic characteristics, self-reported MSK status and diagnosis, healthcare and informal care utilisation, general health status (MEHM), HRQoL (EQ-5D-5L), physical functioning (HAQ-DI) and well-being (ICECAP-A/-O, WHO-5, Happiness VAS). The COnsensus-based Standards for the selection of health status Measurement INstruments (COSMIN) was applied so that we investigated construct validity (clinical, convergent, discriminant validity), internal consistency and reliability [35].

Validity

To assess clinical validity, we investigated whether the MSK-HQ can differentiate between subgroups using Mann–Whitney-U and Kruskal–Wallis tests for comparisons. The association between MSK-HQ score and socio-demographic characteristics were further investigated using ordinary least square (OLS) multiple regression analysis.

To evaluate convergent validity, Spearman's correlation was calculated between MSK-HQ, EQ-5D-5L, EQ VAS, HAQ-DI, ICECAP-A/-O, WHO-5 and Happiness VAS scores. We also investigated correlations between MSK-HQ items, EQ-5D-5L domains and HAQ-DI domains. We expected strong correlation with the 'Mobility' and 'Pain/discomfort' domains of the EQ-5D-5L. The correlations were considered to be strong if the coefficient was over 0.5, moderate between 0.5 and 0.3, and weak under 0.3 [36]. In the multiple regression analysis we also explored associations between MSK-HQ, HRQoL (EQ-5D-5L) and functional status (HAQ-DI) controlling for socio-demographic characteristics. Three models were developed. In the first one, we included the EQ-5D-5L score, in the second one, the HAQ-DI, and in the third model, we included both the EQ-5D-5L and HAQ-DI score to examine how these two scales influence each other's effect. All models were controlled for basic sociodemographic variables (sex, age, education, residency, married/having a partner, living with someone, paid job, income).

EQ-5D-5L index, EQ VAS, ICECAP-A/-O and HAQ-DI scores were calculated for response options of each MSK-HQ items and differences were compared to determine discriminant validity.

Reliability

Internal consistency was examined by Cronbach's alpha (0.7–0.8: acceptable, 0.8–0.9: good > 0.9: excellent) [37]. To assess test–retest reliability in the subsample of 50 participants who have completed the MSK-HQ repeatedly, intraclass correlation coefficient (ICC) was calculated. The ICC can range between 0 (no agreement) and 1 (perfect agreement), indicating the level of agreement (<0.5 –'poor', 0.50–0.749 –'moderate', 0.75–0.900 –'good', > 0.90 –'excellent') [38].

Significance level for all tests was accepted as p < 0.05. Response option of 'I do not know / I do not want to answer' was provided for some items (paid work, income level, MEHM, happinness, EHIS and health care utilisation questions). Respondents who indicated this answer were excluded from the respective analysis of that specific item but their share is provided as 'not reported' among the results. The analysis was performed using IBM SPSS version 25.0 software (IBM Corp., Armonk, NY, USA).

Results

Sample characteristics

Mean (SD) age of the total sample (N=2004) was 48.3 (16.6) years, 1064 (53.1%) respondents were female. Basic socio-demographic characteristics of the sample is presented in Table 1. The mean (SD) MSK-HQ score was 44.13 (9.94), the distribution in the total sample is presented in Fig. 1.

Results on the main outcome measures were as follows (mean, SD): EQ-5D-5L index 0.88 (0.20), EQ VAS 75.8 (20.1), HAQ-DI 0.28 (0.47), ICECAP-A 0.77 (0.19), ICECAP-O 0.83 (0.13), WHO-5 57 (21.5) and Happiness VAS 6.44 (2.30). Results of sample summary statistics are summarised in Additional file 1. Table 1 Sample characteristics and MSK-HQ scores by socio-demographic groups and self-reported health status categories

Variables	Ν	%	MSK-HQ score Mean (SD)
Total	2004	100.0	44.13 (9.94)
Gender			p = .001
Men	940	46.9	44.85 (9.81)
Women	1064	53.1	43.50 (10.01)
Age ^a			p = .000
18–24	192	9.6	47.28 (7.71)
25–34	306	15.3	46.26 (8.29)
35–44	387	19.3	45.90 (8.95)
45–54	332	16.6	43.77 (10.54)
55–64	328	16.4	40.54 (10.87)
65–74	400	20.0	42.97 (10.42)
75+	59	2.9	41.17 (10.15)
Education			p = .000
Primary	608	30.3	42.12 (11.13)
Secondary	968	48.3	44.68 (9.34)
Tertiary	428	21.4	45.77 (8.97)
Settlement type			p = .000
Capital	358	17.9	45.89 (8.88)
Town	1053	52.6	44.26 (9.98)
Village	593	29.6	42.86 (10.29)
Married/having a partner			p = .857
Yes	1300	64 9	44 18 (9 74)
No	704	35.1	44.04 (10.30)
Living with someone in the household	,	0011	p = .224
Yes	1686	84.1	44,24 (9,93)
No	318	15.9	43 59 (10 00)
Paid work			p = .000
Yes	1158	57.8	45 14 (9 21)
No	846	42.2	42 76 (10 71)
Household income category (not reported $N = 304$) ^b	0.10		n = 000
0-718 FUR	752	44 2	μ = .000 41 78 (10 95)
718_1/35 FUR	763	11.2	11.76 (10.93) 11.76 (9.23)
Above 1435 EUR	185	10.9	48 25 (7 27)
Self-perceived health	105	10.5	n = 000
Vervaood	207	10.3	β = .000 51 25 (6 25)
Good	809	10.5	48.63 (6.20)
Fair	784	30.1	41.65 (8.54)
Bad	186	93	29 38 (9 59)
Very bad	18	0.9	20.89 (12.51)
Long standing illness (not reported $N - 90$)	10	0.9	n = 000
	200	47.0	μ — .000 28 55 (7 10)
Yes	1015	53.0	40.26 (10.63)
GALL(not reported N - 32)		0.00	-0.20(10.03)
Severly limited	106	5 /	μ — .000 26 75 (11 27)
Limited hut not severly	500	2.4 20.0	20.73 (11.37)
Not limited	JUU 1776	29.9 61 7	20.24 (7.17) 10 76 (6 70)
NULIIIIIILEU	12/0	04./	40.20 (0.79)

Table 1 (continued)

Variables	Ν	%	MSK-HQ score Mean (SD)
Informal care recipient due to MSK problems in the past three months (not reported $N = 114$)			
Yes	113	6.0	28.41 (9.88)
No	1777	94.0	45.45 (8.87)
Informal caregiver			p = .000
Yes	238	11.9	41.60 (10.69)
No	1766	88.1	44.48 (9.79)

Differences of means were tested by Kruskal-Wallis test and Mann-Whitney U test

GALI Global Activity Limitations

^a There were 55 and 4 participants in age-groups 75–84 and 85–94, respectively

^b Conversion: 1 EUR = 348.35 HUF (the exchange rate in May, 2020)



Clinical validity

Statistically significant differences were found in mean MSK-HQ scores by gender (lower score in women) and age groups (lower score for older participants). Furthermore, higher educational level, living in the capital, having paid work and higher household income, as well as being in a better self-percieved general health status

were also related to significantly higher MSK-HQ mean score. No significant differences were observed by marital status and if the respondent was living alone or with someone in the household. Both informal caregiver and recipient subgroups had significantly lower mean MSK-HQ score. MSK-HQ scores by socio-demographic subgroups and health status are presented in Table 1. MSK-HQ scores were significantly lower if respondents had more serious problems on EQ-5D-5L mobility domain and had difficulties with walking. Differences between subgroups with or without MSK diagnoses were also found to be significant. Furthermore, respondents who had used health care services due to their MSK problems 3 months (outpatient visit) or 1 year (inpatient care) prior to the completion of the survey, had significantly lower scores compared to those who had not. MSK-HQ scores by self-reported MSK health status are presented in Table 2.

Convergent validity

The strongest correlation was observed between MSK-HQ scores and EQ-5D-5L index values, followed by HAQ-DI and EQ VAS (the negative correlation with the HAQ-DI is due to the reverse scoring of the measure). MSK-HQ correlated moderately with ICECAP-A/-O and WHO-5, observed coefficients were fairly similar. However, the correlation with Happiness VAS was only weak. All correlations were found to be significant. Results are presented in Table 3.

In most cases, the association of MSK-HQ items with EQ-5D-5L domains and index scores were strong or moderate, with some exceptions. Weak correlations were observed for 'understanding condition' item of MSK-HQ in all instances. Furthermore, 'anxiety/depression' and 'self-care' domains of EQ-5D-5L were also found to be weakly associated with some of the MSK-HQ items. All correlations were significant. Detailed results are presented in Additional file 2.

HAQ-DI domains showed moderate or strong correlations with most of the MSK-HQ items, except with 'understanding condition' which correlated weakly with all the eight HAQ-DI items and the HAQ-DI score, and with six further items. All correlations were found to be significant. Results are summarized in Additional file 3.

Discriminant validity

Respondents with more severe MSK problems indicated worse states on the outcome measures with some exceptions. Among respondents who reported no problems at all for 'understanding condition' (n=242) and 'confidence in managing' (n=107) MSK-HQ items, a slight increase (better status) was observed in the scores of all scales. Differences between subgroups were found to be significant for all MSK-HQ items and measurement scales. Results are shown in Additional file 4.

Test-retest reliability and internal consistency

All the 50 respondents of the retest subsample answered each question. Among them, the mean MSK-HQ score was 41.30 (\pm 13.79) on the first, and 42.92 (\pm 13.53)

on the second round. The ICC was 0.936 (95% CI: 0.884 – 0.964), indicating excellent level of agreement. Cronbach's alpha was 0.924, indicating good internal consistency.

Regression

Results of the regression analysis are shown in Additional file 5. In models 1 and 2, the EQ-5D-5L and the HAQ-DI were significantly associated with MSK-HQ scores (marginal effects 38.39 and -14.98, respectively). Model 1, which included the sociodemographic background variables and EQ-5D-5L, explained 63% of variance in MSK-HQ score. The performance decreased to 54% for model 2, in which HAQ-DI was included instead of EQ-5D-5L. In model 3, when the two scales were included together, their influence on MSK-HQ remained significant, but with lower marginal effects (27.16 and -6.47, respectively). The best model performance was observed for model 3 which could explain 66% of the variance in MSK-HQ score ($R^2 = 0.663$). For control variables in model 3, significant positive associations were detected with type of residency (living in the capital) and income (above 1435 EUR). Being a women, higher age, primary education and having a partner were found to be negatively associated with MSK-HQ score.

Discussion

We have produced the Hungarian language version of the MSK-HQ measure and tested its psychometric performance in an online cross-sectional survey among the general adult population of Hungary. The tool has been proven to have excellent measurement properties in terms of construct validity, internal consistency, and test-retest reliability. To date, it has been the first MSK-HQ study that was carried out on a representative sample of the general adult population, involving a relatively high sample size compared to previous studies [8, 9, 11, 12, 14, 15, 17, 18]. Furthermore, this has been the first study to assess the association of the MSK-HQ score with EQ-5D-5L and HAQ-DI in a population sample. Another strength of this research is that a number of validated scales were applied to capture aspects of wellbeing alongside the MSK-HQ and their associations were investigated. Using the questions of the Eurostat (EHIS) to assess citizens' self-reported MSK problems and diagnoses provides the first basis for investigating the link between routinely collected national statistics and MSK-HQ scores in further studies.

The new language version of MSK-HQ performed well in our sample as it could differentiate between both socio-demographic groups and self-reported health status categories that were expected to differ in their MSK health. The results of the OLS regression analyses also

Table 2 MSK-HQ scores by musculoskeletal health problems

Variables	Ν	%	Mean (SD)
EQ-5D-5L: Mobility			p=.000
No	1254	62.6	49.06 (6.14)
Slight problems	414	20.7	41.11 (6.61)
Moderate problems	250	12.5	31.23 (8.12)
Severe problems	77	3.8	25.19 (7.89)
Unable to walk	9	0.4	17.89 (14.82)
Difficulty in walking 500 m on level			p=.000
ground without the use of any aid ^a (not reported $N = 18$)			
No difficulty	1527	76.9	47.49 (7.18)
Any difficulty	459	23.1	33.05 (9.78)
Difficulty in walking up or down 12 steps ^a (not reported $N = 13$)			p = .000
No difficulty	1406	70.6	48.21 (6.58)
Any difficulty	585	29.4	34.46 (8.37)
Low back pain or chronic problem ^a (not reported <i>N</i> = 146)			p = .000
No	1108	59.6	48.25 (7.72)
Yes, but not diagnosed	444	23.9	41.73 (8.51)
Diagnosed	98	5.3	39.10 (9.95)
Diagnosed and taking medicine	208	11.2	32.51 (10.56)
Back pain or chronic problem ^a (not reported <i>N</i> = 110)			p=.000
No	1342	70.9	47.14 (8.24)
Yes, but not diagnosed	277	14.6	41.43 (8.61)
Diagnosed	103	5.4	38.87 (9.62)
Diagnosed and taking medicine	172	9.1	30.48 (10.20)
Neck pain or chronic problem ^a			p = .000
(not reported $N = 106$)			
No	1389	73.2	46.85 (8.44)
Yes, but not diagnosed	310	16.3	41.14 (8.91)
Diagnosed	78	4.1	36.15 (10.81)
Diagnosed and taking medicine	121	6.4	30.00 (9.87)
Osteoarthritis of the knee ^a (not reported <i>N</i> = 101)			p=.000
No	1468	77.1	46.49 (8.67)
Yes, but not diagnosed	164	8.6	40.68 (9.13)
Diagnosed	114	6.0	39.77 (9.53)
Diagnosed and taking medicine	157	8.3	32.11 (10.36)
Osteoarthritis of the hip ^a (not reported <i>N</i> =98)			p=.000
No	1602	84.1	46.20 (8.58)
Yes, but not diagnosed	81	4.2	38.95 (11.17)
Diagnosed	93	4.9	36.85 (10.48)
Diagnosed and taking medicine	130	6.8	30.97 (10.00)
Hospital admission in the last 1 year due to MSK health problems (not reported $N = 39$)			p = .000
Yes	235	12.0	36.06 (11.48)
No	1730	88.0	45.35 (9.12)
Seen by specialist in the last 3 months due to MSK health problems (not reported $N = 42$)			p = .000
Yes	322	16.4	36.63 (11.33)
No	1640	83.6	45.79 (8.84)
Seen by family doctor in the last 3 months due to MSK health problems (not reported $N = 34$)			p = .000
Yes	231	11.7	34.87 (10.65)
No	1739	88.3	45.53 (9.02)

Differences of means were tested by Kruskal-Wallis test

^a Questions of the European Health Interview Survey (EHIS) were applied

Table 3 Spearman's correlations of MSK-HQ scores with the EQ-5D-5L index, EQ VAS, HAQ-DI, WHO-5, Happiness VAS, ICECAP-A (age group 18–64) and ICECAP-O (age group 65 +) scores

	MSK-HQ	EQ-5D-5L	EQ VAS	HAQ-DI	ICECAP-A	ICECAP-O	WHO-5 score	Happiness VAS
		index score						
MSK-HQ	1.000	.788	.644	698	.471	.460	.443	.317
EQ-5D-5L index score		1.000	.661	702	.566	.474	.485	.388
EQ VAS			1.000	550	.517	.501	.517	.450
HAQ-DI				1.000	379	427	324	241
ICECAP-A					1.000	_	.628	.606
ICECAP-O						1.000	.606	.579
WHO-5 score							1.000	.621
Happiness VAS								1.000

All correlations are significant at the 0.01 level (2-tailed)

revealed that socio-demographic characteristics, such as gender, age, educational level, marital status, place of residency and income level are significant determinants of MSK-HQ score. Subgroup comparison showed that respondents with self-reported MSK disorders, walking disabilities (based on EHIS questions), prior use of health care services or informal care due to MSK problems had significantly lower MSK-HQ score compared to those who did not report such problems. It can be further concluded that participants with more severe complaints had worse MSK-HQ score. Similar subgroup comparisons have not been published before, therefore our results contribute to the literature by filling this gap.

Our results have further confirmed the strong correlation of MSK-HQ with EQ-5D-5L index, EQ VAS and HAQ-DI that was described in previous studies. In our analysis, we observed a coefficient of 0.788 between MSK-HQ and EQ-5D-5L, which is comparable with results from studies involving MSK patient samples [8, 9, 11, 15]. We would like to emphasise that in our analysis the country-specific Hungarian tariffs were used to calculate the EQ-5D-5L index scores, while at the time of former publications only the value set for England was available [28, 39]. Strong correlation between MSK-HQ and HAQ-DI was observed, with a coefficient of -0.698, although it was lower compared to what has been published in inflammatory arthritis by Norton et al. (r=0.81) [9].

Well-being measurement scales are not commonly used as reference standards, resulting in a scarcity of data within the field of MSK disorders. We found it relevant to broaden the scope and include the WHO-5, ICECAP-A/O and Happiness VAS. Our results have confirmed that the correlation of MSK-HQ with these tools are weaker (but still moderate and significant) compared to that seen with HRQoL (EQ-5D-5L) and physical functioning (HAQ-DI) measures. In previous studies, 'understanding condition' and 'confidence in managing' had weaker relationship with MSK-HQ total score compared to other items [8, 9]. The results of our study further reinforce these observations as we found that 'understanding condition' weakly correlated with all EQ-5D-5L domains. For 'confidence in managing', the correlation was moderate with most of the domains, although coefficients were apparently lower compared to that seen with other items. The same trend was observed for both items in the comparison with HAQ-DI domains.

In our study the Cronbach's alpha of the Hungarian version of MSK-HQ was 0.924, which confirms the good internal consistency of the measure [8, 9, 11, 14, 15]. In addition, the scale was proved to be reliable with an ICC of 0.936, similar to that seen with the Arabic, Turkish and Italian version, although still higher compared to the English, Danish, German, and Norvegian versions [8, 9, 11, 14, 15, 18, 40]. The short time interval between the two assessments partly explains our higher ICC results.

The multiple regression analysis revealed that both EQ-5D-5L and HAQ-DI are significant determinants of MSK-HQ score. The multiple regression analysis also confirmed that MSK-HQ score is significantly assoicated with general HRQoL and phsical functional status. Similar results have been published by Burgess et al., who examined the predictive ability of MSK case-mix adjustment models and found EQ-5D-5L to be significant predictor of 6 months MSK-HQ score [41]. Furthermore, important sociodemographic factors were identified as being significant determinants of MSK-HQ score, such as gender, age, monthly income and educational level. The best predictive ability was observed if both EQ-5D-5L and HAQ-DI were included in the regression analysis along with sociodemographic variables.

Our study has some limitations that must be addressed. Due to the cross-sectional design of our study, we could not assess responsiveness and the minimal important change (MIC). Although, considering previously published MIC results, the results observed by MSK health problems suggest clinically meaningful differences [13, 15, 17]. Inconsistent results were observed for 'understanding condition' and 'confidence in managing' items as an increase was shown in the values of reference scales among those who reported the worst level problems. We further investigated these cases and found that most of these respondents had had no MSK health problems in the last 2 weeks before the completion of the survey, thus these questions were probably not really relevant for them. This issue deserves further consideration (and maybe alternative scoring) for the use of MSK-HQ in non-clinical samples. The survey was conducted online which may cause selection bias. Therefore, caution is needed when using our MSK-HQ results as population norms. Despite the wealth of information acquired from this research, data on effect size, smallest detectable change and MIC are lacking and careful evaluation involving predefined MSK patient groups is still needed. Furthermore, the short time interval between test and retest measures could be a possible source of bias regarding ICC results. However, in previous studies the MSK-HQ was found to have excellent reliability even with lower ICC values and with greater time intervals between the two measures, indicating that these uncertainties probably have an insignificant effect on our conclusion on the reliability of the Hungarian version of the scale.

Conclusions

In summary, we successfully performed the psychometric validation of the Hungarian version MSK-HQ, thus a new language version of the MSK-HQ is now available in Europe. Our study has confirmed the validity of the tool on a representative population sample that broadens the applicability of the MSK-HQ measure. The strong association between MSK-HQ and EQ-5D-5L and HAQ-DI observed in MSK patients has been confimed in this much broader setting. Socio-demographic characteristics have been proved to be significant determinants of MSK-HQ score, which deserves consideration in clinical studies. Moderate but significant associations of MSK-HQ with ICECAP-A/-O, WHO-5 and Happiness VAS measures draws attention to the impact of MSK problems on well-being. The results of our study can be utilized and implemented by clinicians and policy makers as well. The appropriate and aggregated use of MSK-HQ could help to proactively explore MSK health problems of individuals outside the healthcare system, support medical decision making and contribute to develop health policies aiming to improve the HRQoL of patients with MSK disorders.

Abbreviations

COSMIN: COnsensus-based Standards for the selection of health status Measurement INstruments; EHIS: European Health Interview Survey; HAQ-DI: Health Assessment Questionnaire – Disability Index; HRQoL: Health-Related Quality of Life; ICC: Intraclass correlation coefficient; ICECAP-A: ICEpop CAPability measure for Adults; ICECAP-O: ICEpop CAPability measure for Older people; MEHM: Minimum European Health Module; MSK: Musculoskeletal; MSK-HQ: Musculoskeletal Health Questionnaire; OLS: Ordinary Least Square; SD: Standard Deviation; VAS: Visual analog scale; WHO-5: World Health Organisation-Five Well-Being Index.

Supplementary Information

The online version contains supplementary material available at https://doi.org/10.1186/s12891-022-05716-9.

Additional file 1. Sample summary statistics for MSK-HQ, EQ-5D-5L, EQ VAS, HAQ-DI, ICECAP-A/O, WHO-5 and Happiness VAS.

Additional file 2. Spearman correlations of MSK-HQ items with the EQ-5D-5L domains.

Additional file 3. Spearman correlations of MSK-HQ items with the HAQ-DI domains.

Additional file 4. EQ-5D-5L index, EQ VAS, ICECAP-A/O and HAQ-DI scores of respondents on each level of MSK-HQ domains.

Additional file 5. Results of the regression analysis (t-statistics in parentheses).

Acknowledgements

The authors are grateful to Valentin Brodszky (Department of Health Economics, Corvinus University of Budapest, Budapest, Hungary) for his contribution to the research project.

Authors' contributions

All authors contributed to the study conception, design and interpretation of data. Material preparation, data collection and analysis were performed by Áron Hölgyesi, Petra Baji, Miklós Farkas, Ágota Dobos, Zsombor Zrubka, László Gulácsi and Márta Péntek. The first draft of the manuscript was written by Áron Hölgyesi and Márta Péntek and all authors commented on previous versions of the manuscript and revised it critically for important intellectual content. All authors read and approved the final manuscript and agreed to be accountable for all aspects of the work in ensuring that questions related to the accuracy or integrity of any part of the work are appropriately investigated and resolved.

Funding

Open access funding provided by Óbuda University. This research was supported by the Higher Education Institutional Excellence Program of the Ministry of Innovation and Technology in the framework of the 'Financial and Public Services' research project (NKFIH-1163–10/2019) at Corvinus University of Budapest.

Availability of data and materials

The data that support the findings of this study are available from the corresponding author upon reasonable request.

Declarations

Ethics approval and consent to participate

All procedures and methods performed in studies involving human participants were in accordance with the relevant guidelines and regulations, including the ethical standards of the national research committee (Hungarian Medical Research Council) and with the 1964 Helsinki declaration and its later amendments or comparable ethical standards. The protocol of this research has been approved by the Scientific and Ethical Committee of the Medical Research Council of Hungary (no. IV/565–5/2020/EKU). Informed consent was obtained from all individual participants included in the study.

Consent for publication

Not applicable.

Competing interests

In connection with writing this article, Petra Baji, László Gulácsi and Márta Péntek received grant support from the Higher Education Institutional Excellence Program of the Ministry of Innovation and Technology in the framework of the 'Financial and Public Services' research project (TKP2020-IKA-02) at Corvinus University of Budapest. Márta Péntek is member of the EuroQol Group, a not-for-profit organization that develops and distributes instruments that assess and value health. Other authors declare that they have no conflict of interest.

Author details

¹ Doctoral School of Molecular Medicine, Semmelweis University, Budapest, Hungary. ²Department of Health Technology Assessment, National Institute of Pharmacy and Nutrition, Budapest, Hungary. ³Department of Rheumatology, National Institute of Rheumatology and Physiotherapy, Budapest, Hungary. ⁴Department of Health Economics, Corvinus University of Budapest, Budapest, Hungary. ⁵Health Economics Research Center, University Research and Innovation Center, Óbuda University, Budapest, Hungary. ⁶Corvinus Institute for Advanced Studies, Corvinus University of Budapest, Budapest, Hungary. ⁷Department of Accounting and Finance, University of Bristol, Bristol, UK. ⁸Corvinus Center for Foreign Language Education and Research, Corvinus University of Budapest, Budapest, Hungary. ⁹Physiological Controls Research Center, University Research and Innovation Center, Óbuda University, Budapest, Hungary.

Received: 12 May 2022 Accepted: 30 July 2022 Published online: 13 August 2022

References

- Global Burden of Disease Study 2013 Collaborators. Global, regional, and national incidence, prevalence, and years lived with disability for 301 acute and chronic diseases and injuries in 188 countries, 1990–2013: a systematic analysis for the Global Burden of Disease Study 2013. Lancet (London, England). 2015;386(9995):743–800.
- Woolf AD, Erwin J, March L. The need to address the burden of musculoskeletal conditions. Best Pract Res Clin Rheumatol. 2012;26(2):183–224.
- El Miedany Y. Patient reported outcome measures in Rheumatic diseases. Switzerland: Springer International Publishing AG Switzerland; 2016. p. 449.
- Hill JC, Thomas E, Hill S, Foster NE, van der Windt DA. Development and validation of the keele Musculoskeletal Patient Reported Outcome Measure (MSK-PROM). PLoS ONE. 2015;10(4):e0124557.
- Delgado DA, Lambert BS, Boutris N, McCulloch PC, Robbins AB, Moreno MR, et al. Validation of digital visual analog scale pain scoring with a traditional paper-based visual analog scale in adults. J Am Acad Orthop Surg Glob Res Rev. 2018;2(3):e088-e.
- Gossec L, Paternotte S, Aanerud GJ, Balanescu A, Boumpas DT, Carmona L, et al. Finalisation and validation of the rheumatoid arthritis impact of disease score, a patient-derived composite measure of impact of rheumatoid arthritis: a EULAR initiative. Ann Rheum Dis. 2011;70(6):935–42.
- Gossec L, de Wit M, Kiltz U, Braun J, Kalyoncu U, Scrivo R, et al. A patientderived and patient-reported outcome measure for assessing psoriatic arthritis: elaboration and preliminary validation of the Psoriatic Arthritis Impact of Disease (PsAID) questionnaire, a 13-country EULAR initiative. Ann Rheum Dis. 2014;73(6):1012–9.
- Hill JC, Kang S, Benedetto E, Myers H, Blackburn S, Smith S, et al. Development and initial cohort validation of the Arthritis Research UK Musculoskeletal Health Questionnaire (MSK-HQ) for use across musculoskeletal care pathways. Int J Environ Res Public Health. 2016;6(8):e012331.
- 9. Norton S, Ellis B, Santana Suárez B, Schwank S, Fitzpatrick R, Price A, et al. Validation of the Musculoskeletal Health Questionnaire in

inflammatory arthritis: a psychometric evaluation. Rheumatology (Oxford). 2019;58(1):45–51.

- Gibbons E, Fitzpatrick R. An alternative approach to implementing patient-reported outcome measures. Pilot Feasibility Stud. 2018;4(1):96.
- 11. Algarni FS, Alotaibi AN, Altowaijri AM, Al-Sobayel H. Cross-cultural adaptation and validation of the Arabic Version of Musculoskeletal Health Questionnaire (MSK-HQ-Ar). Int J Environ Res Public Health. 2020;17(14):5168
- Akkubak Y, AnaforoĞlu KÜlÜnkoĞlu B. Reliability and validity of the Turkish Version of Arthritis Research UK Musculoskeletal Health Questionnaire. Arch Rheumatol. 2020;35(2):155–62.
- Price AJ, Ogollah R, Kang S, Hay E, Barker KL, Benedetto E, et al. Determining responsiveness and meaningful changes for the Musculoskeletal Health Questionnaire (MSK-HQ) for use across musculoskeletal care pathways. BMJ Open. 2019;9(10):e025357.
- Galeoto G, Piepoli V, Ciccone E, Mollica R, Federici C, Magnifica F, et al. Musculoskeletal health questionnaire: translation, cultural adaptation and validation of the Italian version (MSK-HQ-I). Muscles Ligaments Tendons J. 2019;9(2):295–303.
- Karstens S, Christiansen DH, Brinkmann M, Hahm M, Mc CG, Hill JC, et al. German translation, cross-cultural adaptation and validation of the Musculoskeletal Health Questionnaire: a cohort study. Arch Rheumatol. 2020;56(6):771–9.
- Christiansen DH, McCray G, Winding TN, Andersen JH, Nielsen KJ, Karstens S, et al. Measurement properties of the musculoskeletal health questionnaire (MSK-HQ): a between country comparison. Health Qual Life Outcomes. 2020;18(1):200.
- Scott DIC, McCray DG, Lancaster PG, Foster PNE, Hill DJC. Validation of the Musculoskeletal Health Questionnaire (MSK-HQ) in primary care patients with musculoskeletal pain. Semin Arthritis Rheum. 2020;50(5):813–20.
- Tingulstad A, Van Tulder MW, Rysstad T, Tveter AT, Hill JC, Grotle M. Validity and reliability of the Norwegian version of the Musculoskeletal Health Questionnaire in people on sick leave. Health Qual Life Outcomes. 2021;19(1):191.
- 19. Riva S, Pravettoni G. Value-Based model: a new perspective in medical decision-making. Front Public Health. 2016;4:118
- Fautrel B, Alten R, Kirkham B, de la Torre I, Durand F, Barry J, et al. Call for action: how to improve use of patient-reported outcomes to guide clinical decision making in rheumatoid arthritis. Rheumatol Int. 2018;38(6):935–47.
- Proud L, McLoughlin C, Kinghorn P. ICECAP-O, the current state of play: a systematic review of studies reporting the psychometric properties and use of the instrument over the decade since its publication. Qual Life Res. 2019;28(6):1429–39.
- Versteegh M, Knies S, Brouwer W. From good to better: new dutch guidelines for economic evaluations in healthcare. Pharmacoeconomics. 2016;34(11):1071–4.
- Hicks S, Tinkler L, Allin P. Measuring subjective well-being and its potential role in policy: perspectives from the UK Office for National Statistics. Soc Indic Res. 2013;114(1):73–86.
- 24. (NICE) NIfHaCE. Developing NICE guidelines: the manual. 2014. Available from: https://www.nice.org.uk/process/pmg20/resources/developing-nice-guidelines-the-manual-pdf-72286708700869.
- Office HCS. Population Census 2011. Available from: http://www.ksh.hu/ nepszamlalas/tables_regional_00?lang=en.
- Cox B, van Oyen H, Cambois E, Jagger C, le Roy S, Robine JM, et al. The reliability of the minimum European health module. Int J Public Health. 2009;54(2):55–60.
- Herdman M, Gudex C, Lloyd A, Janssen M, Kind P, Parkin D, et al. Development and preliminary testing of the new five-level version of EQ-5D (EQ-5D-5L). Qual Life Res. 2011;20(10):1727–36.
- Rencz F, Brodszky V, Gulácsi L, Golicki D, Ruzsa G, Pickard AS, et al. Parallel valuation of the EQ-5D-3L and EQ-5D-5L by time trade-off in Hungary. Value Health. 2020;23(9):1235–45.
- 29. Fries JF, Spitz P, Kraines RG, Holman HR. Measurement of patient outcome in arthritis. Arthritis Rheum. 1980;23(2):137–45.
- Al-Janabi H, Flynn TN, Coast J. Development of a self-report measure of capability wellbeing for adults: the ICECAP-A. Qual Life Res. 2012;21(1):167–76.
- Coast J, Flynn TN, Natarajan L, Sproston K, Lewis J, Louviere JJ, et al. Valuing the ICECAP capability index for older people. Soc Sci Med. 2008;67(5):874–82.

- Baji P, Farkas M, Dobos Á, Zrubka Z, Gulácsi L, Brodszky V, et al. Capability of well-being: validation of the Hungarian version of the ICECAP-A and ICECAP-O questionnaires and population normative data. Qual Life Res. 2020;29(10):2863–74.
- Flynn TN, Huynh E, Peters TJ, Al-Janabi H, Clemens S, Moody A, et al. Scoring the lcecap-a capability instrument. Estimation of a UK general population tariff. Health Econ. 2015;24(3):258–69.
- Topp C, Østergaard S, Søndergaard S, Bech P. The WHO-5 well-being index: a systematic review of the literature. Psychother Psychosom. 2015;84:167–76.
- Prinsen CAC, Mokkink LB, Bouter LM, Alonso J, Patrick DL, de Vet HCW, et al. COSMIN guideline for systematic reviews of patient-reported outcome measures. Qual Life Res. 2018;27(5):1147–57.
- Cohen J. Set correlation and contingency tables. Appl Psychol Meas. 1988;12(4):425–34.
- Bland JM, Altman DG. Statistics notes: Cronbach's alpha. BMJ. 1997;314(7080):572.
- Koo TK, Li MY. A guideline of selecting and reporting intraclass correlation coefficients for reliability research. J Chiropr Med. 2016;15(2):155–63.
- 39. Devlin NJ, Shah KK. Valuing health-related quality of life: An EQ-5D-5L value set for England. 2018;27(1):7–22.
- Akkubak Y, AnaforoĞluKÜlÜnkoĞlu B. Reliability and validity of the Turkish Version of Arthritis Research UK Musculoskeletal Health Questionnaire. Arch Rheumatol. 2020;35(2):155–62.
- Burgess R, Lewis M, Hill JC. Musculoskeletal case-mix adjustment in a UK primary/community care cohort: Testing musculoskeletal models to make recommendations in this setting. Musculoskelet Sci Pract. 2021;56:102455.

Publisher's Note

Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.

Ready to submit your research? Choose BMC and benefit from:

- fast, convenient online submission
- thorough peer review by experienced researchers in your field
- rapid publication on acceptance
- support for research data, including large and complex data types
- gold Open Access which fosters wider collaboration and increased citations
- maximum visibility for your research: over 100M website views per year

At BMC, research is always in progress.

Learn more biomedcentral.com/submissions

